Author's response to reviews

Title: Comparative cost-efficiency of the EVOTECH Endoscope Cleaner and Reprocessor versus manual cleaning plus automated endoscope reprocessing in a real-world Canadian hospital endoscopy setting.

Authors:

Lindy Forte (lindy.forte@sympatico.ca)
Cynthia Shum (cshum@its.jnj.com)

Version: 5 Date: 8 August 2011

Author's response to reviews: see over
August 8, 2011

Editor
BioMed Central Ltd
Floor 6
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear Editor,

Please find enclosed our revised manuscript entitled “Comparative cost-efficiency of the EVOTECH Endoscope Cleaner and Reprocessor versus manual cleaning plus automated endoscope reprocessing in a real-world Canadian hospital endoscopy setting”.

Thank you for the further review of our manuscript. We appreciate the first referee’s comments that the changes made to date have improved the quality of the manuscript.

As per the recommendation of Referee 1, we have made the final compulsory revision by adding a footnote to Table 7 to clarify for the reader that this table does not include the initial acquisition price of either device.

We hope that the revised manuscript will meet with the Editors’ approval and the standards for publication in the Journal. Please do not hesitate to email or call me if you have any further questions or comments. I can be reached at 416-453-8350 or lindy.forte@sympatico.ca.

Sincerely,

Lindy Forte, MSc